Author's response to reviews

Title: Effects of Ischemic Preconditioning On Ischemia/Reperfusion-Induced Arrhythmias by Upregulation of Connexin 43 Expression

Authors:

Zhenguang Chen (zhenguangchen@tom.com)
Honghe Luo (hhluo@126.com)
Mei Zhuang (mzhuang@tom.com)
Lie Cai (liecai@yahoo.com.cn)
Chunhua Su (such@tom.com)
Yiyan Lei (leiyy@sina.com)
Jianyong Zou (jianyongz@sohu.com)

Version: 3 Date: 26 May 2011

Author's response to reviews: see over
Dear Editor,

Thank you very much for sending me the proof manuscript. The revisions were made as follows.

(1) page 4, line 7 from the bottom: "with the approval of the local ethics committee" was added.
(2) Figures 1(a), (b), (c), (d) were emerged and the figure was uploaded as Fig1.
(3) We removed all highlighting and tracking from the manuscript.
(4) References: all authors were added.
(5) Figures 1 and 2 were re-drawn.

Thank you again.

Sincerely,

Zhenguang Chen